

#### **OPEN ACCESS**

EDITED BY Zeeshan Fareed, Huzhou University, China

REVIEWED BY
Tomiwa Sunday Adebayo,
Cyprus International University, Türkiye
Shah Fahad,
Leshan Normal University, China

\*CORRESPONDENCE Rinat A. Zhanbayev, zhanbayevrinat@gmail.com Muhammad Irfan, irfansahar@bit.edu.cn

SPECIALTY SECTION
This article was submitted
to Environmental
Economics and Management,
a section of the journal
Frontiers in Environmental Science

RECEIVED 05 September 2022 ACCEPTED 11 November 2022 PUBLISHED 05 January 2023

#### CITATION

Zhanbayev RA, Yerkin AY, Shutaleva AV, Irfan M, Gabelashvili K, Temirbaeva GR, Chazova IY and Abdykadyrkyzy R (2023), State asset management paradigm in the quasi-public sector and environmental sustainability: Insights from the Republic of Kazakhstan. *Front. Environ. Sci.* 10:1037023. doi: 10.3389/fenys.2022.1037023

### COPYRIGHT

© 2023 Zhanbayev, Yerkin, Shutaleva, Irfan, Gabelashvili, Temirbaeva, Chazova and Abdykadyrkyzy. This is an openaccess article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

# State asset management paradigm in the quasi-public sector and environmental sustainability: Insights from the Republic of Kazakhstan

Rinat A. Zhanbayev (b) 1\*, Albina Y. Yerkin (b) 2, Anna V. Shutaleva (b) 3,4, Muhammad Irfan (b) 5,6,7\*, Kakhaberi Gabelashvili<sup>8</sup>, G. R. Temirbaeva (b) 9, Irina Yu. Chazova<sup>10</sup> and Rimma Abdykadyrkyzy<sup>11</sup>

<sup>1</sup>National Engineering Academy of the Republic of Kazakhstan, Almaty, Kazakhstan, <sup>2</sup> Al-Farabi Kazakh National University, High School of Economics and Business, Almaty, Kazakhstan, <sup>3</sup>Department of Philosophy, Ural Federal University named after the first President of Russia B. N. Yeltsin, Ekaterinburg, Russia, <sup>4</sup> Department of Social and Humanitarian Disciplines, Ural State Law University named after V. F. Yakovlev, Ekaterinburg, Russia, <sup>5</sup>School of Management and Economics, Beijing Institute of Technology, Beijing, China, <sup>6</sup>Center for Energy and Environmental Policy Research, Beijing Institute of Technology, Beijing, China, <sup>7</sup>Department of Business Administration, Ilma University, Karachi, Pakistan, <sup>8</sup>Almaty University of Power Engineering and Telecommunications named after Gumarbek Daukeev, Almaty, Kazakhstan, <sup>9</sup>Zhezkazgan University named after O. A. Baykonurova, Zhezkazgan, Kazakhstan, <sup>10</sup>FGBOU VO Udmurt State University, Izhevsk, Russia, <sup>11</sup>Abai Kazakh National Pedagogical University, Almaty, Kazakhstan

This study aims to improve state regulation and administration of quasi-state sector of environmental sustainability in the Republic of Kazakhstan within the social sustainability paradigm. This study is due to the need to search the balance of business interests, and efficient use of resources and their conservation on a global scale. The issues of state asset management are identified and ways to resolve them are proposed based on the corporate foresight methodology using the strategic planning mechanism. As a result of the analysis of the tools introduced into the practice of leading transnational corporations, the main directions are identified in which it is advisable to take measures aimed at streamlining and optimizing the quasi-public sector in the Republic of Kazakhstan. This article presents proposals for improving state regulation and administration of quasi-public companies in the Republic of Kazakhstan. The outcomes of this study can assist policymakers, experts, and stakeholders in gaining awareness about these problems while simultaneously improving sustainability practices.

### KEYWORDS

clean energy transition, environmental sustainability, strategic planning, state assets, quasi-public sector

### 1 Introduction

Today, more than ever, the problem of balancing the interests of business, both public and private, and the global problems of scarcity of resources and the related problem of their preservation on a global and regional scale is acute. In international practice, the sustainable development of the state is presented as interconnected with the quasi-state sphere, which needs a balance of economic, social, and environmental spheres (Francis, 2001; Kim, 2003; Collier, 2005). The quasi-public sector in any state is the foundation of the country's economy, on which all modern processes are based, both in terms of institutional affiliation and in terms of sectoral structure. Since the state's assets are a rather complex and multi-level system, their state management is considered a rather multifaceted process.

The main components of sustainable development are political, social, cultural, economic, and environmental sustainability. In a market economy, the achievement of sustainable development is interrelated with environmental sustainability. The reason is the increase in resource efficiency use through the introduction of more advanced and environmentally friendly (non-waste) technologies becoming the main task of many states, including quasi-state companies and market participants. Currently, the relationship between macroeconomic and environmental parameters is given great importance (Shan et al., 2021; Ojekemi et al., 2022). In this regard, scientifically based environmental management, recycling of production, and consumption of wastes are relevant topics (Adebayo, 2022a; Adebayo, 2022b; Du et al., 2022). The reason is that ecological production, at this stage of development of the economy of states, is becoming an increasingly significant factor in competitiveness. The production of ecological products and waste processing are promising types of entrepreneurial activity (Adebayo et al., 2022; Akadiri et al., 2022; Hossain et al., 2022). The higher the environmental sustainability of market participants, the higher the environmental efficiency of the development and functioning of the economy, and the social and environmental stability of society.

However, quasi-public companies have a greater impact on the economy than small private companies. Therefore, quasi-public companies can increase the level of environmental awareness and activity in society by resorting to formal and informal education methods from childhood, as well as regular posting of information on official Internet resources and social media accounts, and other measures. Populations living in environmentally sensitive areas should have greater opportunities to participate in the decision-making process that will affect society. Environmental sustainability issues should be reflected in educational programs and in all strategic decisions, the consequences of which are subject to regular evaluation (Shutaleva et al., 2020a; García-Morís and Martínez-MedinaTrainee, 2022; Ma and Jin, 2022). The

proposed measures will make it possible to form an "ecological culture of the population" in society, and the promotion of environmental knowledge—the development of environmental education and enlightenment to ensure sustainable development. Thus, environmental sustainability is the effective management of natural resources in production activities, allowing them to be preserved for future needs. And quasi-state companies and state companies cannot dispose of it as an inexhaustible source of resources, even if their goal is to satisfy human needs through consumption.

This article is devoted to studying ways to improve state regulation and administration of quasi-state companies within the sustainability paradigm in the Republic of Kazakhstan. Currently, the Government of the Republic of Kazakhstan is asking a difficult task to reform the quasi-public sector. Economic and social problems in the Republic of Kazakhstan demonstrate the need to search for effective strategies for the development of a quasi-public sector. This circumstance determines the relevance of the presented study of the existing situation, considering the social, economic, and environmental aspects of the serial development of society.

One of the main problems of the modern development of the Republic of Kazakhstan is the need to determine a clear role and place of the subjects of the quasi-public sector in the system of state planning. This situation distinguishes the state of the Republic of Kazakhstan from developed countries and creates the need to comprehend international experience and apply its methodology, considering the specifics of the socio-economic development of the country.

In the Budget Code of the Republic of Kazakhstan, the concept of "state assets" is set out in the following wording: "state assets are property, and non-property benefits and rights that have a valuation received into state ownership because of past operations or events" (Budget code of the Republic of Kazakhstan, 2022). Quasi-state companies have more liquid assets than the state companies. However, quasi-state companies play an important role in the markets (Makushkin and Lapshin, 2021). The issue is aggravated by the accelerating processes of digitalization of the process of interaction between subjects of various forms of ownership. The Internet determines the life of modern society, including the main business processes (Afonasova et al., 2019; Khachaturyan, 2021; Nicolaou, 2021). It is necessary to consider the specifics of the digital economy and the need for state regulation of electronic interaction. In the Republic of Kazakhstan, this trend also applies to public sector procurement and information resources (Subjects of the quasipublic sector, 2022). Currently, the relevance of improving asset management efficiency is only increasing.

The Republic of Kazakhstan strives to be consistent with the goals and principles of sustainable development. However, several important aspects should be considered when analyzing the specifics of the implementation of the concept of sustainability in the Republic of Kazakhstan. Such aspects

include structural, eco-nomic-geographical, and traditional priorities of the Kazakh society.

The interaction between the public and private sectors has historical roots in the Republic of Kazakhstan. The modern trend of this interaction is to shift the focus on the problem of sustainable development of cities and the growth of the quality of life. These processes require a rethinking of the goals, institutions, and mechanisms of policy in the field of public asset management. We proceed from the fact that the policy of managing state assets should achieve systemic and transparent management of state property and ensure social and economic stability in the country and its regions.

The economy of the Republic of Kazakhstan is characterized by low dynamics in the development of the private sector and dependence on primary industries (Aubakirova, 2020; Nurpeisova et al., 2021). This circumstance predetermines the fixation of the quasi-public sector as an essential factor in the development of the economy. Currently, the Government of the Republic of Kazakhstan is taking certain actions to strengthen the management of the quasi-public sector (Kazakh Government Adopts Plan to Reform Quasi-Public Sector, 2022). These measures include the separation of the role of the state as owner and manager, the creation of national holdings for the management of state assets, and development institutions. However, today there are challenges in the economy and social problems in the population of the regions. This circumstance allows a state that the quasi-public sector's current model has several systemic shortcomings.

A legitimate mechanism for the effective management of state assets in the quasi-public sector of the economy of Kazakhstan is impossible without a comprehensive analysis of the state of management of state assets. We believe that the key problem is in the inefficient use and management of state assets, even if they can be sold commercially. This trend leads to the lack of proper influence of state assets on economic growth in the region and improvement of the quality of life of the population of the Republic of Kazakhstan. In this regard, the authors of this article intend to propose the results of a study aimed at creating an effective system for managing state assets in the quasi-public sector.

This article is aimed to find ways to improve state regulation and administration of quasi-state companies in the Republic of Kazakhstan within the framework of the sustainability paradigm. The relevance of this study is in the need to introduce balanced strategic planning, rethinking the goals and mechanisms of socio-economic state policy in the Republic of Kazakhstan. We proceed from the international theory and practice of applying the methodology of corporate foresight using the strategic planning mechanism. We undertook a study to develop recommendations that will allow reformatting the management system of the quasi-public sector in the Republic of Kazakhstan. The implementation of this goal determined the following logic of this study. Section 2 shows the results of the

study of existing approaches and issues of quasi-public sector development in the context of the sustainability paradigm. Section 3 presents the foresight methodology as the main one for the proposed study. Section 4 presents an analysis of the process of regulating the activities of quasi-state companies with the participation of government departments. The quality of regulation of quasi-state companies needs to be addressed the most problematic issues, among which we have considered the following: lack of a clear institutional framework; low financial efficiency of quasi-state companies; high share of the state in the economy; lack of a comprehensive system for monitoring and evaluating the activities of quasi-state companies; the imperfection of a single information resource about quasistate companies; weak corporate governance. Section 5 presents proposals for improving state regulation and administration of quasi-public companies in the Republic of Kazakhstan. Section 6 presents conclusions drawn about the ways to streamline and optimize the quasi-public sector in the Republic of Kazakhstan.

### 2 Literature review

Sustainability is a paradigm that affects the environmental, economic, technological, industrial, social, and other areas of society in its current and future development perspective. In 1987, the United Nations General Assembly published the report Our Common Future, which presented the concept of sustainability as a concept for the development of society, involving the responsible use of natural resources, considering the preservation of nature for the life of future generations [(WCED, 1987), p. 43]. One of the main conclusions of the World Conference in Rio de Janeiro (1992) is the recognition that the existing production and consumption systems will inevitably lead to a global catastrophe (Rio Declaration on Environment and Development, 1992). This concern and the formation of a paradigm of sustainability are based on the fact of environmental destruction and the catastrophe that threatens all of humanity. In this regard, innovations in the sphere of macroeconomic and environmental parameters (Shan et al., 2021; Ojekemi et al., 2022), production and economy (Schaltegger et al., 2012; Baimuratov et al., 2020; Yizhe et al., 2020), environmental management, recycling of production (Adebayo, 2022a; Adebayo, 2022b; Du et al., 2022), environmental ecology (Frazier et al., 2019; Valko, 2021a; Leigh and Li, 2015; Piscitelli and D'Uggento, 2022), production of ecological products and waste processing (Adebayo et al., 2022; Akadiri et al., 2022; Hossain et al., 2022), urban development (Trencher and Karvonen, 2017; Krupkin and Gorodnova, 2018; Dixon and Brears, 2021), education for sustainable development (Berchin et al., 2012; Shutaleva et al., 2020b; Valko, 2021b), improvement of the welfare of the population (Pissourios, 2013; Mendoza-Cavazos and Leal Filho, 2019; Shutaleva et al., 2022), preservation

of cultural traditions, and mutual respect of members of society (Alaimo, 2018; Loginov, 2019; Bakeeva and Biricheva, 2021).

For the study of state regulation and administration of quasistate companies, the concept of economic sustainability as the main element of sustainability is of particular interest. Economic sustainability is fundamental to social sustainability, which requires a basic level of financial resources to meet the needs of society's members (Dernbach, 1998; Zeijl-Rozema et al., 2008; Rai et al., 2021). The economic perspective of sustainable development emphasizes the maintenance of various types of capital (Kevin, 2010; Costanza, 2020; Porreca, 2020). Sustainability is a synergistic phenomenon; that is, it is formed in the interaction of economic, environmental, and social spheres and types of capital (Illankoon et al., 2017; Eklova, 2020; Simangan et al., 2021). However, the social, environmental, and economic aspects of sustainability tend to be mutually exclusive in practice, which creates threats for Morelli (2011) paid attention to such an aspect of economic sustainability as the inclusion in it of an analysis to minimize social costs and compliance with environmental asset protection standards.

One of the significant aspects of economic sustainability is the development of a unified measure of economic and environmental aspects of sustainability. Scott (2013) defined economic sustainability as long-term competitiveness, profitability, and shareholder satisfaction. Competitiveness and profitability must be associated with innovativeness, resource optimization, elimination of short-term thinking, and environmental hazards of production.

Economic sustainability needs a balance of efficiency and sustainability, it is determined by the nature and measurable structures of economic systems (Lietaer et al., 2009). Economic efficiency assumes that the requirement to satisfy more needs at the expense of fewer resources is met. The sustainable functioning of firms implies financial viability and the protection of the firm from external and internal threats in the present and future perspective (Naciti, 2019; Rustam et al., 2020; Hermundsdottir and Aspelund, 2021). Sustainability and security are the most important components of the enterprise economy, ensuring its integrity and unity as a system. An organization's security involves protecting its human and intellectual potential, information, technology, capital, and profits. The company's economic security is realized through the effective use of corporate resources to ensure the stable functioning of the organization. In this regard, the strategic development of economic, organizational, legal, engineering, and social measures is particularly important. Strategic planning allows the company to remain in a state of life support in the face of economic crisis and risk (Hastings, 2000; Lagâri et al., 2021; Waiganjo et al., 2021).

In the macroeconomic aspect, the following criteria of economic sustainability are significant: competitiveness (Buhalis, 2000; Armenski et al., 2018; Arunachalam and

Fountis, 2021), innovativeness (Hemmelskamp et al., 2000; Jänicke et al., 2000; Akbar, 2021), and public debt (Albu and Albu, 2021; Grosu et al., 2021; Adeve and Karabou, 2022). In 2005, Spangenberg (2005) noted that in the context of sustainable development, such criteria of economic sustainability as inflation and trade imbalance are practically not discussed. However, in studies of recent years (Barthélemy and Cléaud, 2018; Elimam, 2021; Skvortsova et al., 2022), trade balance, imbalance, and inflation are the subjects of research. Spangenberg (2005) also notes that traditional economic sustainability measures, such as aggregate demand and consumption levels are secondary in the current debate. However, these issues are significant for the modern discussion of economic sustainability (Sağlam and Egeli, 2018; Aydın and Tirkolaee, 2022; Chan et al., 2022).

One of the significant topics of the sustainable development of society is the liberal intellectual tradition and its criticism in modern studies (Hart-Landsberg and Burkett, 2005; Petrucciani, 2020; Sørensen, 2022). The reason for the criticism is that the transition to the market has led to the widespread dissemination of quasi-public and quasi-private phenomena based on the interpenetration between the state and society. Francis (2001) points to an alternative concept implemented in China's emerging market economy.

Developing the quasi-public sector of the economy is important for the sustainable development of society and the state (Francis, 2001; Qian et al., 2019; Kamenev, 2021). At the same time, studies of the functioning of the quasi-public sector in the Republic of Kazakhstan relate to certain problems, for example, the mechanism for monitoring procurement in the quasi-public sector (Khamitov, 2020), institutional and investment mechanisms to support the agro-industrial complex (Taubayev et al., 2017). Issues of the quasi-public sector in the Republic of Kazakhstan need further research for further development and improvement.

The state has a large mass of assets and corporate governance tools. In connection with this circumstance, there are opinions about the need to establish control over the activities of state corporations (Avdasheva and Simachev, 2009). One of the significant problems is the problem of inefficient management of state property and the contradiction in the performance of enterprises in public and private sectors (Fominykh, 2004; Luzan, 2004; Zhavoronkov, 2004).

The development of an effective state policy for the prudent use of regional assets is necessary to reduce the degree of the negative impact of international sanctions on the state. At the same time, the main role of state policy should not be in the distribution of income between regions but in creating conditions for the involvement of these assets in economic activity, which should improve the population's quality of life (Emelyanov et al., 2016). At the same time, tools for assessing the effectiveness of state property management are not properly involved in different regions of the country (Kamaev, 2012). Solovyov (2008) believes that when managing state assets of the economy, it is important

to observe the principle of ensuring a balance of interests between economic efficiency and social responsibility.

The key indicator of the effectiveness of public administration is its impact on the population's quality of life (Baimuratov et al., 2020; Vasiliev and Sushko, 2021; Zhanbayev et al., 2021). The ability to apply country experience and, in general, international trends in public asset management in "transition" countries depends on many factors that influence the development of these countries and specific national and international forces (So, 2005; Roje and Redmayne, 2021). Nazif Çatık et al. (2020) analyzed asset pricing models considering changes in oil prices and exchange rates. They refine the methodology for assessing asset management's effectiveness in terms of market prices' influence on them and conclude that this factor affects industry profitability less than the exchange rate return.

Yan et al. (2020) investigate the incentive effect of reputation and the deterrent effect of oversight on the allocation of public assets in budget management. The researchers conclude that the incentive effect of reputation is effective for a high cost of the budget unit, and the constraining effect of supervision is effective for a budget unit with a low cost.

It should be noted that a significant part of the research in the field of state property management states inefficiency due to bureaucracy and lack of accountability of state organizations. At the same time, research on the activities of private structures is devoted, on the contrary, to the accumulation of profits from the investments of private investors, and the lack of analysis of the effectiveness of public investments (Noring, 2019).

### 3 Materials and methods

Methodological database of research in the field of public asset management. When studying the state of private asset management, empirical general scientific methods of cognition were also used. A method of observing and collecting facts of regularities in the processes of reproduction in their nature. This method of identifying the neonatal state of state assets in a quasistate republic was determined in Kazakhstan and revealed a problem.

A systematic approach is used as a methodological approach, within which methods of logic, statistical analysis, and synthesis are provided. This methodological approach served by our procedure for the corporate foresight mechanism to use the right of transnational corporations (Kováříková et al., 2017; Semke and Tiberius, 2020; Wenzel, 2022).

Foresight is based on the method of self-expression of expert assessments and includes the active formation of an image of the future and the determination of development priorities. This method makes it possible to develop a forward-looking program of action to respond to the main requirements and achieve the goal. The obvious advantages of corporate foresight have served

to form improved expectations from its practical application, which is associated with the choice of research methodology. There was an opinion that now there is a transition from management to strategic foresight (Gatignon and Xuereb, 1997; Danielson, 2014). However, it is believed that the possibilities of foresight are still limited. Foresight cannot fully replace the selection system, but it can significantly increase its effectiveness as its complement.

We find with those authors who believe that foresight's strength is not in tools and methods but in influencing the minds and views of people. Carries out foresight competencies that can transform their organizational culture (Ansoff et al., 1976; Ruff, 2015). Thus, using the foresight mechanism attracts more responsibility for improving the efficiency of quasi-state litigation. When addressing issues of research methodology, emphasis is placed on the approach to private asset management from an objective orientation towards dynamically changing emerging markets in the context of special transformations of the competitive landscape. The implementation of this goal is possible when using the foresight methodology. The effectiveness of such a methodology is its benchmark for achieving sustainable economic development. Of course, the foresight mechanism is crucial for transnational businesses (Bereznoy, 2017). For us, foresight is of interest in sequential analysis in the next protocol for the economy of Kazakhstan, namely, in the quasi-statistical protocol.

We proceed from the fact that using foresight tools makes it possible to increase the efficiency of managing third-party assets. The application of this methodology is carried out in the main directions aimed at streamlining and optimizing the quasi-energy sectors in Kazakhstan: Firstly, the need to distinguish between three levels of the quasi-public sector, depending on the degree of influence on operational activities and the loss of income for the development of the country's economy. Secondly, creating a new national holding showed that the attitude toward the state is strategic and a priority for the country's economy. Thirdly, the creation of a share of the quasi-public sector increases the risk of reducing the functions of the state structure and body to use the synergistic effect of public-private consumption. Finally, the quasi-public sector should not restrict competition but oppose the stimulation of business development in all its forms and the diversification of the economy.

### 4 Results

We justified the application of the triple helix model in forming a conceptual mechanism aimed at the interaction of the prominent participants: business, government, education, and science, which correlates with (Baimuratov et al., 2020). In this study, we believe it is correct to use and test the methodology of the above-mentioned conceptual approach for

TABLE 1 Elements of the corporate foresight mechanism for quasi-public companies.

No.	Tools of the mechanism	Content	Main participants
	Assessment of the current state of affairs	Formation of an information basis for the analysis of the past, present, and future situations of interaction between business, society, quasi-state companies, and the state	Quasi-state companies
	Mission	Definition of a comprehensive goal that includes both internal and external guidelines for cooperation	Quasi-state companies
	The general goal	Establishment of specific end states of the interaction model or the desired result	Quasi-state companies
	SWOT analysis	Analysis of the strengths and weaknesses of the socio-economic development of the region, as well as potential opportunities and threats $\frac{1}{2}$	Quasi-state companies
	Priorities		
	Business development	A set of actions aimed at implementing favorable conditions for effective development that meet the needs of society and business	Business community and quasi-state companies, government
	Cooperation	Development of joint projects of stakeholders and businesses on the so-called "information mediation."	Business community and quasi-state companies, government
	Clustering	Development of joint development programs within the framework of the creation of industry groups $% \left( 1\right) =\left( 1\right) \left( 1$	Business community and quasi-state companies, government
	Partnership in moving forward	Ensuring cooperation with stakeholders to create a favorable market environment	Business community and quasi-state companies, government
	Practice and Discourses (Planning)	Practice and discourses in which the forecasting and design of the "Future in the present" is carried out	Business community and quasi-state companies, government
	Shaping the Future (Implementation	Development of a strategy for the implementation of the concept	Business community and quasi-state companies, government
	of Strategic Directions)	Formation of scenarios, drawing up roadmaps that increase competitiveness and quickly respond to market changes in the internal and external environment when making managerial business decisions	

public-private interaction based on the corporate foresight toolkit. Table 1 presents elements of corporate foresight for the sector on which the sustainable development of the economy depends, which is the quasi-public sector.

Here are the main elements of the corporate foresight mechanism for quasi-state companies that most strongly affect the achievement of sustainable economic development goals. The expansion of the activities of quasi-state companies requires the development of a single vector of development, considering the specifics of the market segment, the application of a systematic approach to monitoring compliance with the norms of optimal conditions for assessing their economic and social results, and considering the development priorities of the Republic of Kazakhstan. At the same time, special attention should be paid to effective infrastructural support for business and the identification of stakeholders, raising the level of awareness of business and society, and identifying the country's resource and market opportunities. In this regard, the application element can be used to increase the use of digital technologies.

The lack of a clear institutional framework for quasi-state companies is noteworthy, which negatively affects the ability to determine priority goals and objectives and to measure their achievements qualitatively and quantitatively. Clustering allows tracking changes in the current architecture of the system of state and quasi-state companies. On this basis, it is possible to activate

the system of monitoring and evaluating their effectiveness while influencing the goal of development stability.

Action planning involves solving the problem of financial returns from the activities of quasi-public companies and, consequently, state participation. To this end, an action plan is being drawn up, and ongoing measures are aimed at modernizing the activities of state and quasi-state companies, including increasing investment attractiveness.

From our point of view, the results of the search for the results of activities of quasi-state companies are especially relevant for the economy of the Kazakhstani regions. Today, the issues of public-private ownership in the regions are in a state of lack of demand. This tendency is due to the low growth of the innovative dynamics of the socio-economic development of the regions and the subsequent weak competitive environment. In this plane, the role of the public sector represented by the quasi-public sector, which, as you know, manages the distribution of most of the resources, is more important than ever. Thus, planning their activities is a necessary factor in the economy's stability.

Sustainable development is possible with the balance of three main components: the growth of economic and social responsibility and environmental balance. In this article, the emphasis was on economic growth and social things, and environmental things were considered in our observations.

TABLE 2 Information on the number of state institutions in the Republic of Kazakhstan (for 2021).

No.	Name	Quantity, units
	State bodies (thousand)	8.3
	State institutions in the field of education (schools, out-of-school organizations), (thousand)	7.8
	Objects of culture and sports (houses of culture, sports schools, archives, libraries), (thousand)	1
	Objects in the field of healthcare and social protection (hospitals, clinics, centers for adaptation, employment)	500
	Institutions are represented in the structure of the ministries of defense, internal affairs (military units, detention centers, penitentiary institutions)	400
	Youth centers	180
	Other institutions (service centers, veterinary stations, transport organizations)	150
	Total, (thousand)	18.3

TABLE 3 State and quasi-state companies in the non-social sphere of the Republic of Kazakhstan.

No.	Name	Quantity (units)
	Organizations of the Samruk-Kazyna Fund	204
	Veterinary stations	194
	Water supply	79
	Development Institute	63
	Organizations supporting the activities of state bodies	37
	Mass media organizations (media)	33
	Organizations in the field of housing and communal services (heat, energy, gas supply, water supply, landscaping)	25
	Investment funds	29
	Single operators	20
	Organizations in the field of energy and electricity supply	23
	Plants	22
	Transport organizations	13
	State monopolies	11
	Service companies	29
	APK	7
	National companies	10
	FEZ operator	2
	Other	17
	Total	818

From this point of view, we examined the analysis of the process of regulating the activities of quasi-state companies with the participation of government departments. This study made it possible to identify the most problematic issues that can

sufficiently improve the quality of their management, and this should be focused on the sustainable development of the economy.

### 4.1 The first problem is the lack of a clear institutional framework

In 2021, the state will be the owner of about 25 thousand state institutions, state enterprises, JSCs, and LLPs. 18.3 thousand (Table 2).

6.4 thousand subjects of the quasi-public sector, including five thousand organizations (or 78%) are social (education (3.5 thousand), culture and sports (more than 800), healthcare and social protection (more than 700); related to the non-social sphere, 818 organizations, excluding organizations included in the Comprehensive Privatization Plan (736 organizations) (Table 3).

So, out of 25 thousand state institutions, state enterprises, joint-stock companies, and limited liability partnerships, 23.4 thousand organizations belong to the social sphere, these are state bodies, organizations in the field of education, healthcare, social protection, culture, and sports.

These subjects strategically influence the economic system at the state level. However, objectively, such a mass of subjects needs to create a balanced mechanism for managing quasi-state structures. The reason is the lack of consistency based on their classification and accounting makes it difficult to develop a single vector for the development of the industry. It is necessary to identify the main groups of quasi-state companies, in respect of which it is necessary to apply various management mechanisms. To complete this task, the following factors must be considered:

- 1. First, the system does not divide quasi-state companies into commercial and non-commercial activities. As a result, the same approach is applied to all enterprises; common requirements are imposed, even though organizations are all different in terms of goals and objectives of creation. For example, national companies face different challenges than schools and hospitals. There are also non-commercial enterprises among the national companies—Kazakhinvest JSC, by its nature, cannot generate income; therefore, it cannot be put on a par with other commercial joint-stock companies of the quasi-public sector.
- 2. Lack of systematic approaches to defining organizational and legal forms. As a result, when creating quasi-state companies, the issue of applying one or another type of organizational and legal forms is determined by the initiating state body itself. For example, in the field of education, organizations operate in various organizational and legal forms, which is associated with the lack of a unified approach to their financing. Thus, schools are registered as state institutions or state enterprises

on the right of economic management. This is because, for state-owned enterprises, the per capita funding standard is higher than for state institutions. As a result, this leads to a different amount of funding, the level of remuneration of teachers, as well as accounting for the property.

At the same time, it is not entirely clear what institutional differences exist between one or another status of quasi-state companies: national management holdings, national holdings and companies, and national and regional development institutions.

- 3. Accounting for state property is carried out at four levels. Each of these levels is vested with the right to independently determine the procedure for managing state property, which creates conditions for a fragmented policy in this area. Accounting in the authorized body (State Property and Privatization Committee) is not subject to the property of organizations, the founder, participant, or shareholder. In this regard, completeness and reliability of accounting are not ensured, as there are cases of lack of information about individual objects.
- 4. The lack of a clear institutional framework for quasi-state companies negatively affects the system for monitoring and evaluating the effectiveness of quasi-state companies, the ability to determine the necessary goals and objectives based on the specifics of the activities of quasi-state companies, measuring their achievements and implementation.

# 4.2 The second problem is the low financial efficiency of quasi-state companies

Net income at the end of 2020 amounted to 67,98,510 thousand tenges. That is, it remained at the level of 2019 for 67,91,000 thousand tenges. At the same time, 95% of the net income of quasi-state companies falls on the share of NWF Samruk-Kazyna JSC and only 5% on the share of other legal entities. In 2020, the total income of quasi-state companies for the reporting period amounted to 23.1 billion tenges, which is 15% lower compared to 2019 and amounted to 27.08 billion tenges. The total expenses in 2020 of quasi-state companies for the reporting period amounted to 14.2 billion tenges, which is 26% lower than the same period in 2019, which amounted to 19.2 billion tenges.

The practice of state financing of quasi-state companies continues. So, in 2019, 817.4 billion tenges was allocated from the republican budget, in 2018—659.2 billion, and 2017—838.8 billion tenge. In addition, funds allocated from the budget to support businesses in various sectors of the economy have become a source of income for many quasi-state companies. For example, receiving 0.5%–1% from the

budget or the National Fund, some quasi-state companies finance businesses and citizens at inflated rates of 9%.

The level of external debt of the quasi-public sector remains quite high and continues to grow. In 2020, its size amounted to 20.1 billion US dollars. There are systemic shortcomings in the dividend policy. At the end of 2020, the growth of debt obligations compared to the level of 2019 amounted to 15%. The amount of dividends paid to the state does not exceed 7% of the net income received. This fact is associated with an increase in the unprofitability of quasi-state companies and other distributions in favor of the shareholder to finance non-core facilities. Thus, strengthening control over the debt load and clear regulation of corporate and dividend policy should be one of the stages of the implemented strategic planning.

### 4.3 The third problem is that the state's share in the economy remains high

The share of state participation in the economy remains high compared with the Organization for Economic Cooperation and Development indicators. Thus, the actual share of the consolidated assets of the national managing holdings of Kazakhstan to the country's GDP amounted to more than 50%. The main tools to reduce the state's share in the economy are privatization and a reduction in the list of activities carried out by quasi-state companies. However, the potential for using these mechanisms is insufficient. For example, the selection of quasi-state companies for privatization is carried out without an appropriate level of transparency and indepth analysis of the consequences of privatization. In addition, despite the reduction in the list of permitted activities for quasi-state companies in 2020 by 107 types, 306 activities currently remain, some of which can be performed in the private sector. The reduction did not lead to the actual optimization of quasi-state companies since the reduced activities were not related to their main activities.

According to the list of activities approved by Government Decree No. 1095 of 28 December 2015, quasi-public sector entities can still carry out activities in the areas of interpretation and translation, buying and selling real estate, cleaning activities, web portals, bus transportation, advertising in the media, the activities of security organizations and other areas. The imposition of a moratorium on the creation of new quasi-state companies did not help increase efficiency.

# 4.4 The fourth problem is the lack of a comprehensive system for monitoring and evaluating the performance of quasi-state companies

The applied system for monitoring and evaluating the effectiveness of state property management does not allow for improving the quality of the activities of quasi-state companies.

We note the following shortcomings:

 Only certain categories of quasi-state companies are subject to assessment. Mainly these are state institutions and enterprises. Large quasi-state companies (for example, NWF Samruk-Kazyna JSC and its portfolio companies) are not subject to the requirements for assessing the effectiveness of their activities. A significant share of quasi-state companies in terms of assets and budgetary investments remains without regular objective assessment.

2. For the assessment, the Rules for monitoring the effectiveness of state property management, approved by the Government Decree of 4 December 2012, No. 1546, as well as the Rules for assessing the effectiveness of state property management, approved by order of the Ministry of National Economy of 11 March 2015 No. 193, are used. However, they do not allow to fully carry out monitoring of the activities of quasi-state companies. Thus, the Rules for the implementation of monitoring (1,546) do not regulate the mechanism for implementing recommendations based on monitoring results, which reduces the effectiveness of the analytical process. The rules for evaluating the effectiveness of state property management (193) also do not reflect the real situation of the activities of quasistate companies. They do not contribute to improving the efficiency of state property management. The reason is that the assessment uses common parameters. For example, criteria are used for "Reducing the number of organizations with a negative financial result," "The presence or absence of stateowned enterprises and state-controlled joint-stock companies (LLPs) that do not have key indicators that can be quantified in the approved development plan."

# 4.5 The fifth problem is the imperfection of a single information resource on quasistate companies

Currently, the informational secrecy of quasi-state companies is maintained. In the public domain, there is no data on the structure of many companies, financial activities, staffing, dividends paid to the shareholder, and other aspects. This trend is especially true for state-owned enterprises that do not publish the necessary information about their activities.

By transferring part of the functions of information technologies of portfolio companies to a competitive environment, several tasks can be solved: 1) development of the information technology management function, 2) ensuring business continuity, taking into account the balance between obtaining benefits and optimizing risks and resources, 3) developing the domestic market information technologies.

In addition, it is also important to ensure the information security of quasi-state companies. To build a system for managing and ensuring information security at a group level according to the principle of a service model, the following actions are required: 1) increase the level of maturity of management and information security processes, 2) reduce their risks, 3) reduce response time, 4) reduce the costs of eliminating the consequences of information security incidents. Thus, the key factor for the success of quasi-state companies should be the effective completion of the digital transformation processes of quasi-state companies.

### 4.6 The sixth problem is weak corporate governance

Corporate governance in Kazakhstan, not only in quasistate companies, is used quite formally and is not in demand by business owners, shareholders, and stakeholders. At the same time, in joint-stock companies, the issue of corporate governance is regulated by the Law of the Republic of Kazakhstan "On Joint Stock Companies" and other regulations. However, only separate corporate governance elements have been implemented for state-owned enterprises and limited liability partnerships. The current legal framework does not provide for the responsibility of members of the supervisory and executive bodies of quasi-state companies, which, in turn, affects their effectiveness. Issues of transparency and accountability of quasi-state companies have not been settled. In terms of decision-making, there is a risk of the excessive influence of the central and local executive bodies in charge of the relevant industry, their excessive interference in the activities of quasi-state companies, including the centralization of tasks, often at odds with the strategy of quasi-state companies.

The sustainable state's economic basis is provided to a large extent by the quasi-public sector. Therefore, there is no doubt about the need to determine priorities in the management of quasi-state companies that would give a vector for the sustainable development of the economy. In this regard, we assume the following recommendations for reforming the quasi-public sector management system in the Republic of Kazakhstan.

- 1. It is necessary to form a new institutional structure for the activities of quasi-state companies, reflecting a clear delineation of functions, processes, tasks, and criteria for assessing their effectiveness.
- Considering the classification and type of activity, for each group of quasi-state companies, establish separate financial criteria and indicators for the implementation of strategic and tactical tasks.
- 3. Improve the financial discipline of quasi-state companies.
- 4. Reduce the share of state participation in the economy.
- 5. Ensure transparency of activities and openness of quasi-state companies of the Republic of Kazakhstan.

6. Increase the information content of monitoring and evaluation of quasi-state companies.

7. Ensure the quality of corporate governance in accordance with global ethical standards.

### 5 Discussion

This study is based on the analysis of problems and the search for their solutions in the regulation of the activities of quasi-state companies. In this regard, in this section we discuss proposals for improving state regulation and administration of quasi-state companies. In this section, we discuss such aspects as economic growth, the social dimension, and environmental issues of the proposed recommendations. The legitimacy of this approach lies in the fact that the quasi-public sector, like no other, is most represented in the social sphere and the environment.

# 5.1 It is necessary to form a new institutional structure for the activities of quasi-state companies, reflecting a clear delineation of functions, processes, tasks, and criteria for assessing their effectiveness

It is important to revise the current classification of quasistate companies based on the goals and objectives of these organizations. First, it is important to provide for the separation of legal entities with the participation of the state, depending on the commercial and non-commercial orientation. Further, commercial, and non-commercial quasi-state companies are proposed to be divided into two types presented in Table 4:

# 5.2 Considering the classification and type of activity for each group of quasi-state companies, establish separate financial criteria and indicators for implementing strategic and tactical tasks

If the organizations are non-profit social and implement national policy, then the following conditions should apply to them:

- they should not be subject to requirements in terms of profitability, while their activities should be break-even;
- indicators of their effectiveness should be the quality provision of social services and the achievement of strategic (key) indicators;
- financing of these groups should be carried out by the estimated principle and through the state order.

To commercial infrastructure, strategic quasi-state companies, as well as development institutions:

- there must be requirements for profitability;
- indicators of their effectiveness to consolidate the achievement of strategic indicators;
- implementation of financing at the expense of own funds and through budget financing and borrowing in the capital market.

# 5.3 Improve the financial discipline of quasi-state companies

We consider the experience of some developed countries (for example, Australia), where methodological instructions (Commonwealth Government Business Enterprises – Governance and Oversight Guidelines) have been developed for government business companies that help manage their activities. The instructions contain companies' main financial, strategic, and reporting requirements to comply with. By analogy with these instructions, it is necessary to develop a document that fixes the basic requirements for the activities of quasi-state companies.

The risk of insolvency of quasi-state companies can become a threat to the stability of the country's economy. In this regard, it is necessary to strengthen monitoring of the level of external debt of quasi-state companies. For example, constant monitoring and control over external and internal loans of the CSC have been carried out since 2018 in pursuance of the order of the Head of the Presidential Administration "On approval of the Action Plan for the implementation of the instructions of the President of the Republic of Kazakhstan given at the opening of the third session of the Parliament of the Republic of Kazakhstan of the sixth convocation on 4 September 2017." The issue of maximizing the payment of dividends is being worked out as part of the instructions of the Head of State following the expanded meeting of the Government of 10 July 2020. Transparency and accountability of the activities of the quasi-public sector are being worked out within the framework of the National Action Plan for the implementation of the Address of the Head of State.

# 5.4 Reduce the share of state participation in the economy

It is advisable to carry out separate work to identify and liquidate or restructure or privatize inefficient (unprofitable) organizations. The list and goals of the activity of quasi-state companies should be reviewed, considering new strategic tasks and national priorities of the Republic of Kazakhstan.

TABLE 4 Commercial and non-commercial quasi-public companies.

### No. Commercial quasi-public companies

Infrastructural, strategic

JSC "National Company "Kazakhstantemirzholy"; JSC National Atomic Company Kazatomprom; JSC "National Company "KazMunayGas"; National Welfare Fund Samruk-Kazyna Development institutions JSC Baiterek National Managing Holding JSC; JSC "Development Bank of Kazakhstan", Socialentrepreneurial corporations

Non-profit quasi-state companies Social non-profit quasi-state companies

Companies implementing national policy on specific issues

In addition, conducting a critical assessment of their activities and the return on budgetary investments is necessary. For example, it is advisable to consider the consolidation of development institutions in the development and export promotion field. The same work can be done with organizations in the field of technological development. For example, following the Government Decree dated 30 July 2016 No. 450, Export Insurance Company KazakhExport JSC, National Company KAZAKH INVEST JSC, Kazakhstan Industry and Export Center QazIndustry JSC are designated as institutions in the field of development and promotion of exports. QazTrade Trade Policy Development Center JSC. Also, the issues of supporting exporters are dealt with by the Development Bank of Kazakhstan JSC, Damu JSC, the Foreign Trade Chamber of NCE Atameken, and animats.

Technological development issues are assigned to the Center for Engineering and Technology Transfer JSC, QazTechVentures JSC, and Astana Innovations JSC. Considering the low efficiency of social entrepreneurial corporations and diversion to non-core activities, it is necessary to take measures to reduce the portfolio of social entrepreneurial corporations, their subsidiaries, and affiliates by transferring them to a competitive environment, as well as to get rid of non-core assets, including them in the list of privatized objects. For example, an analysis of the activities of this institution conducted by the Ministry of National Economy shows that among 17 social and entrepreneurial corporations, there is no unified approach to the implementation of state programs and territorial development programs. Often their activities do not correspond to the main goal, namely, to support business initiatives, stimulate economic activity in the growth points of the regions, and are unprofitable. The return on assets of social entrepreneurial corporations in most regions has negative values. More than a third of the subsidiaries of social and entrepreneurial corporations are inactive. There is a lack of activity in these enterprises in the search for investment opportunities.

# 5.5 Ensure the transparency of activities and openness of quasi-state companies of the Republic of Kazakhstan

We propose the following measures to ensure the disclosure of information about quasi-state companies:

- use of modern information technologies to improve the efficiency of collection, processing, analysis, and public disclosure of information by issuers and other financial market participants, as well as to provide access to databases for all interested parties;
- it is necessary to introduce the standard of openness of quasi-state companies.

For the most effective satisfaction of users' information requests regarding the activities of quasi-state companies, it is necessary to create a data access infrastructure that provides for the organization of feedback. Creating effective feedback will allow the organization's management to quickly respond to changing information user requests, which will increase the organization's value in the eyes of its stakeholders and create a competitive advantage.

# 5.6 Increase the information content of monitoring and evaluation of quasi-state companies

It is important to revise the approach to accounting for state property. At the same time, the Register of State Enterprises and Institutions, Legal Entities with State Participation in the Authorized Capital should be updated. In addition, it is necessary to ensure at the legislative level that the requirements for monitoring and evaluating the effectiveness of state property management and financial and economic activities are extended to all quasi-state companies. It is also important to strengthen the Rules for monitoring the effectiveness of state property management (1,546) by regulating the mechanism of execution based on the results of monitoring recommendations. Include the Accounts Committee in the system for monitoring the effectiveness of state property management. During the revision of the Rules for assessing the effectiveness of state property management (193), the emphasis will be shifted to the socio-economic efficiency of quasi-state companies.

# 5.7 Ensure the quality of corporate governance following global ethical standards

To improve corporate governance, ensure the inclusion of more qualified independent directors on the boards of directors and the exclusion of formal members of the board of directors. Strengthen the board of directors and its members' responsibility for implementing strategic objectives. To increase human resources, provide for the phased introduction of minimum qualification requirements for senior positions and the Board of Directors of quasi-state companies (for example, professional certification in corporate governance, FRM, CFA, ACCA). Maintaining a register of highly qualified independent directors and strengthening their role in making management decisions is necessary.

Review the relationship of the quasi-public sector with government bodies, considering OECD standards. As one of the measures in this direction, we propose the exclusion of civil servants from among the members of the Board of Directors. Implementing a set of proposals will create conditions for forming a highly efficient, compact, and transparent quasi-public sector.

### 6 Conclusion and policy propositions

We investigated the tools, methods, and algorithms for the administrative management of state assets, and qualitative analysis of the advantages and disadvantages of the Republic of Kazakhstan quasi-state companies. The new role and functions of the state in a globalizing world require a rethinking of the goals, institutions, and mechanisms of socio-economic state policy based on the introduction of balanced strategic planning, in which the main indicator is the achievement of sustainable development. The basic factor is the sphere of administrative and strategic management of quasi-state companies.

The quasi-public sector of the Republic of Kazakhstan has several shortcomings the lack of a clear institutional framework for activities, the low financial efficiency of entities, and a high level of the state in the economy. In this regard, the main measures are proposed to reformat the management system of the quasi-public sector to focus it on the sustainable development of the economy as a whole:

- the transformation of the institutional structure and target orientation of quasi-state companies;
- development and approval of requirements taking into account the type of activity of quasi-state companies with a classification by profitability, objectives, strategic indicators, and funding procedures;
- control and financial discipline of quasi-state companies with a decrease in state participation;
- transparency to internal and external stakeholders;
- modernization of indicators for monitoring and evaluating quasi-state companies;

- an updated model of corporate governance in those organizations where it is applied.

As a result of the analysis of the tools introduced into the practice of leading transnational corporations, the main directions were identified in which it is advisable to take measures aimed at streamlining and optimizing the quasipublic sector in the Republic of Kazakhstan:

- clearly distinguish between three levels of quasi-state companies, depending on the degree of intervention in the operating activities of subsidiaries and affiliates, as well as on the significance of assets for the development of the country's economy;
- quasi-state companies should ensure the symbiosis of protecting the national interests of the public sector of the Republic of Kazakhstan and the corporate efficiency of the commercial sector;
- the creation of quasi-state companies is justified if their activities are not aimed at duplicating the functions of a state body but at increasing the synergistic effect of publicprivate partnerships;
- quasi-state companies should not narrow the competitive environment but provide a multiplier effect for the development of small and medium-sized businesses and related sectors of the economy.

This article examined the effectiveness of managing state assets of the quasi-public sector of the economy in the Republic of Kazakhstan. We proceed from the international theory and practice of applying the corporate foresight methodology using the strategic planning mechanism. However, the study is focused on the current situation in the Republic of Kazakhstan. Today, the economy of the Republic of Kazakhstan is faced with the task of getting out of the low dynamics of the development of the private sector and overcoming dependence on the primary industries. In connection with this circumstance, the development of the quasi-public sector is considered by the government of the republic as the most important factor in the development of the economy. However, the challenges in the economy and the social problems of the population of the regions of the Republic of Kazakhstan show that the current model of the quasi-public sector has several systemic shortcomings. Therefore, we have made recommendations for the development of the quasi-public sector refer to the Republic of Kazakhstan. The new institutional structure of quasi-state companies' activities should reflect a clear delineation of functions, processes, tasks, and criteria for evaluating their effectiveness. Improving the efficiency of quasi-state companies is directly related to their discipline in the financial system. Therefore, it is necessary to develop financial criteria and classification of indicators for the implementation of strategic and tactical tasks. It is impossible to reduce the share of state participation in the economy without

mechanisms to ensure transparency of activities and openness of quasi-state companies of the Republic of Kazakhstan. Therefore, the main principles should be information openness and ethical standards of public administration and the activities of the quasi-public sector.

### Data availability statement

The original contributions presented in the study are included in the article/supplementary material, further inquiries can be directed to the corresponding authors.

### **Author contributions**

Conceptualization, RZ, MI, and AS; Methodology, RZ; Validation, AY, KG, MI, and GT; Formal analysis, IC; Investigation, RZ; Resources, RZ; Data curation, RZ and AS; Writing—original draft preparation, RZ; Writing—review and editing, RZ and AS; Visualization, AY; Supervision, RZ and AS; Project administration, RZ. All authors have read and agreed to the published version of the manuscript.

### References

Adebayo, T. S., Bekun, F. V., Rjoub, H., Agboola, M. O., Agyekum, E. B., and Gyamfi, B. A. (2022). Asymmetric effect of structural change and renewable energy consumption on carbon emissions: Designing an SDG framework for Turkey. *Environ. Dev. Sustain.* 2022, 1–29. doi:10.1007/s10668-021-02065-w

Adebayo, T. S. (2022). Environmental consequences of fossil fuel in Spain amidst renewable energy consumption: A new insights from the wavelet-based granger causality approach. *Int. J. Sustain. Dev. World Ecol.* 29, 579–592. doi:10.1080/13504509.2022.2054877

Adebayo, T. S. (2022). Renewable energy consumption and environmental sustainability in Canada: Does political stability make a difference? *Environ. Sci. Pollut. Res.* 29, 61307–61322. doi:10.1007/s11356-022-20008-4

Adeve, K. A., and Karabou, E. F. (2022). Public debt and development sustainability issues in the west african economic and monetary union (WAEMU). Cogent Econ. Finance 10, 1. doi:10.1080/23322039.2022.2079177

Afonasova, M., Panfilova, E., Galichkina, M. A., and Ślusarczyk, B. (2019). Digitalization in economy and innovation: The effect on social and economic processes. *Pol. J. Manag. Stud.* 19 (2), 22–32. doi:10.17512/pjms.2019.19.2.02

Akadiri, S. S., Adebayo, T. S., Nakorji, M., Mwakapwa, W., Inusa, E. M., and Izuchukwu, O. (2022). Impacts of globalization and energy consumption on environmental degradation: What is the way forward to achieving environmental sustainability targets in Nigeria? *Environ. Sci. Pollut. Res.* 29, 60426–60439. doi:10.1007/s11356-022-20180-7

Akbar, K. (2021). How economic sustainability is affected by innovation performance and sustainable manufacturing. World J. Adv. Res. Rev. 11 (01), 247–255. doi:10.30574/wjarr.2021.11.1.0350

Alaimo, L. S. (2018). Sustainable development and national differences: An European cross-national analysis of economic sustainability. RIEDS-Rivista Italiana di Econ. Demografia e Statistica-Italian Rev. Econ. Demogr. Stat 72 (3), 101–123. Avaliable At: http://www.sieds.it/listing/RePEc/journl/2018LXXII\_N3\_RIEDS\_09\_20\_Alaimo\_ok.pdf.

Albu, A.-C., and Albu, L.-L. (2021). Public debt and economic growth in Euro area countries. A wavelet approach. *Technol. Econ. Dev. Econ.* 27 (3), 602–625. doi:10.3846/tede.2021.14241

### **Funding**

This study was funded and supported by the Science Committee of the Ministry of Education and Science of the Kazakhstan No. AP13068164 Development of tools aimed at modeling socioeconomic systems for sustainable development of society.

### Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

### Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Ansoff, I., Declerck, R. P., and Hayes, R. L. (1976). From strategic planning to strategic management. London, New York: Wiley.

Armenski, T., Dwyer, L., and Pavluković, V. (2018). Destination competitiveness: Public and private sector tourism management in Serbia. *J. Travel Res.* 57, 384–398. doi:10.1177/0047287517692445

Arunachalam, V., and Fountis, A. (2021). A research study on the Sustainable Competitiveness and the Macro Economic Factors of the IT Sector in India which contributes to sustainability and the rise of social development. *J. Contemp. Issues Bus. Gov.* 27, 1548–1556. doi:10.47750/cibg.2021.27.03.205

Aubakirova, G. M. (2020). Transformational change in the economy of Kazakhstan. Stud. Russ. Econ. Dev. 31, 113–119. doi:10.1134/S1075700720010037

Avdasheva, S., and Simachev, Yu. (2009). State corporations: Is it possible to evaluate corporate governance? *Vopr. Ekon.* 6, 97–110. doi:10.32609/0042-8736-2009-6-97-110

Aydın, N. S., and Tirkolaee, E. B. (2022). A systematic review of aggregate production planning literature with an outlook for sustainability and circularity. *Environ. Dev. Sustain.* 2022. 1–42. doi:10.1007/s10668-022-02304-8

Baimuratov, U. B., Zhanbayev, R. A., and Sagintayeva, S. S. (2020). The triple helix model for the conceptual mechanism of cooperation between higher education and business: The regional aspect. *EoR.* 16, 1046–1060. doi:10.17059/ekon.reg.2020-4-3

Bakeeva, E. V., and Biricheva, E. V. (2021). I" and collective responsibility. *Vestnik SPbSU. Philosophy Confl. Stud.* 37, 41–52. doi:10.21638/spbu17.2021.104

Barthélemy, J., and Cléaud, G. (2018). Trade balance and inflation fluctuations in the euro area.  $Macroecon.\ Dyn.\ 22$  (4), 931–960. doi:10.1017/S1365100516000456

Berchin, I. I., Dutra, A. R. A., and de Andrade Guerra, J. B. S. O. (2012). How do higher education institutions promote sustainable development? A literature review. *Sustain. Dev.* 27 (4), 1204–1222. doi:10.1002/sd.2219

Bereznoy, A. (2017). Corporate foresight in multinational business strategies. Foresight STI Gov. 11 (1), 9–22. doi:10.17323/2500-2597.2017.1.9.22

Budget code of the Republic of Kazakhstan (2022). Code of the republic of Kazakhstan. Available At: https://online.zakon.kz/Document/?doc\_id= 30364477&pos=5;-108#pos=5;-108 (accessed on May 18, 2022).

Buhalis, D. (2000). Marketing the competitive destination of the future. *Tour. Manag.* 21 (1), 97–116. doi:10.1016/S0261-5177(99)00095-3

Chan, I., Franks, B., and Hayek, M. N. (2022). The 'sustainability gap' of US broiler chicken production: Trade-offs between welfare, land use and consumption. *R. Soc. open Sci.* 9, 210478. doi:10.1098/rsos.210478

Collier, P. M. (2005). Governance and the quasi-public organization: A case study of social housing. *Crit. Perspect. Account.* 16, 929–949. doi:10.1016/j.cpa.2004.

Costanza, R. (2020). Valuing natural capital and ecosystem services toward the goals of efficiency, fairness, and sustainability. *Ecosyst. Serv.* 43, 101096. doi:10. 1016/j.ecoser.2020.101096

Danielson, M. R. (2014). The impact of corporate foresight and strategic orientation on performance. Denmark: Aarhus University: Aarhus.

Dernbach, J. C. (1998). Sustainable development as a framework for national governance. *Case West. Reserve Law Rev.* 49 (1), 1–105. Available online: https://ssrn.com/abstract=1089413 (accessed on May 18, 2022).

Dixon, T. J. (2021). "City visions: Toward smart and sustainable urban futures," in *The palgrave encyclopedia of urban and regional futures*. Editor R. Brears (Cham: London, UK: Palgrave Macmillan). doi:10.1007/978-3-030-51812-7\_14-1

Du, L., Jiang, H., Adebayo, T. S., Awosusi, A. A., and Razzaq, A. (2022). Asymmetric effects of high-tech industry and renewable energy on consumption-based carbon emissions in MINT countries. *Renew. Energy* 196, 1269–1280. doi:10.1016/j.renene.2022.07.028

Eklova, K. (2020). Sustainability of buildings: Environmental, economic and social pillars. *Bus. IT X* (2), 2–11. doi:10.14311/bit.2020.03.01

Elimam, H. (2021). Impact of carbon taxes, economic growth, globalization, forest rent, inflation and urbanization on sustainable development in China. *Int. J. Res. Granthaalayah.* 9 (12), 303–314. doi:10.29121/granthaalayah.v9.i12.2021. 4450

Emelyanov, S. G., Vertakova, Yu.V., and Solodukhina, O. I. (2016). Methodological approaches to the development of state policy in the field of rational use of regional assets. *Proc. Southwest. State Univ.* 6, 102–110.

Fominykh, A. (2004). Relative inefficiency of the state and private sectors: The statistical approach. *Vopr. Ekon.* 9, 64–75. doi:10.32609/0042-8736-2004-9-64-75

Francis, C.-B. (2001). Quasi-public, quasi-private trends in emerging market economies: The case of China. *Comp. Polit.* 33 (3), 275–294. doi:10.2307/422404

Frazier, A. E., Bryan, B. A., Buyantuev, A., Chen, L., Echeverria, C., Jia, P., et al. (2019). Ecological civilization: Perspectives from landscape ecology and landscape sustainability science. *Landsc. Ecol.* 34, 1–8. doi:10.1007/s10980-019-00772-4

García-Morís, R., and Martínez-MedinaTrainee, R. (2022). Trainee teachers' perceptions of socio-environmental problems for curriculum development. *Soc. Sci.* 11 (10), 445. doi:10.3390/socsci11100445

Gatignon, H., and Xuereb, J. (1997). Strategic orientation of the firm and new product performance. J. Mark. Res. 34, 77–90. doi:10.1177/002224379703400107

Grosu, A. C., Pintilescu, C., and Zugravu, B. (2021). Trends in public debt sustainability in Central and Eastern EU countries. *Post-Communist Econ.* 34, 173–195. doi:10.1080/14631377.2020.1867431

Hart-Landsberg, M., and Burkett, P. (2005). China and socialism: Market reforms and class struggle. New York, USA: Monthly Review Press.

Hastings, M. (2000). Sustainable development: The challenge of transition. Cambridge, USA: Cambridge University Press. doi:10.1017/CBO9780511536021

Hemmelskamp, J. (2000). "Environmental taxes and standards: An empirical analysis of the impact on innovation," in *Innovation-oriented environmental regulation. ZEW economic studies.* Editors J. Hemmelskamp, K. Rennings, and F. Leone (Heidelberg, Germany: Spinger). doi:10.1007/978-3-662-12069-9\_15

Hermundsdottir, F., and Aspelund, A. (2021). Sustainability innovations and firm competitiveness: A review. *J. Clean. Prod.* 280, 124715. doi:10.1016/j.jclepro.2020.

Hossain, M. E., Rej, S., Hossain, M. R., Bandyopadhyay, A., Tama, R. A., and Ullah, A. (2022). Energy mix with technological innovation to abate carbon emission: Fresh evidence from Mexico applying wavelet tools and spectral causality. *Environ. Sci. Pollut. Res. Int.* 2022, 1–22. doi:10.1007/s11356-022-2355-2

Illankoon, I. C., Tam, V. W., and Le, K. N. (2017). Environmental, economic, and social parameters in international green building rating tools. *J. Prof. Issues Eng. Educ. Pract.* 143, 05016010. doi:10.1061/(ASCE)EI.1943-5541.0000313

Jänicke, M., Blazejczak, J., Edler, D., and Hemmelskamp, J. (2000). "Environmental policy and innovation: An international comparison of policy frameworks and innovation effects," in *Innovation-oriented environmental regulation. ZEW economic studies.* Editors J. Hemmelskamp, K. Rennings, and F. Leone (Heidelberg, Germany: Spinger). doi:10.1007/978-3-662-12069-9-7

Kamaev, R. A. (2012). To the question of the effectiveness of the use of state property as a component of the infrastructure of the regions. *Bull. Volgograd State Tech. Univ.* 14 (103), 161–167.

Kamenev, G. (2021). "Optimal control in agent-principle problems for the quasi-public sector," in 14th International Conference Management of large-scale system development (MLSD), Moscow, Russian Federation, 27-29 September 2021 (IEEE). doi:10.1109/MLSD52249.2021.9600215

Kazakh Government Adopts Plan to Reform Quasi-Public Sector (2022). Increase economic growth. Available At: https://inbusiness.kz/ru/news/kazakhgovernment-adopts-plan-to-reform-quasi-public-sector-increase-economic-growth-37711 (accessed on May 18, 2022).

Kevin, O. (2010). Strategic intellectual capital management in multinational organizations: Sustainability and successful implications. Hershey, PA: IGI Global.

Khachaturyan, A. (2021). Human capital in the digital economy. Resour. Environ. Econ. 4 (1), 314–324. doi:10.25082/REE.2022.01.002

Khamitov, Z. M. (2020). The analysis and monitoring of the current state of procurement in the quasi-public sector of the Republic of Kazakhstan. jour.~3, 240–245. doi:10.46914/1562-2959-2020-1-3-240-245

Kim, J. (2003). The emergence of the quasi-government sector in korea: Some policy implications. *Int. Rev. Public Adm.* 8 (1), 115–129. doi:10.1080/12294659. 2003.10805022

Kováříková, L., Grosová, S., and Baran, D. (2017). Critical factors impacting the adoption of foresight by companies. Foresight (Colch). 19 (6), 541-558. doi:10.1108/FS-02-2017-0009

Krupkin, A., and Gorodnova, N. (2018). Development of the smart city concept in sustainable economy. *IOP Conf. Ser. Mat. Sci. Eng.* 365 (2), 022056. doi:10.1088/1757-899X/365/2/022056

Lagâri, E. C., Erzurumi, A. M., and Tomur, B. F. (2021). Strategic planning and crisis management in textile sector in Turkey. *J. Strategic Manag.* 5 (4), 1–12. doi:10. 53819/81018102t4023

Leigh, M. C., and Li, X. (2015). Industrial ecology, industrial symbiosis and supply chain environmental sustainability: A case study of a large UK distributor. *J. Clean. Prod.* 106, 632–643. doi:10.1016/j.jclepro.2014.09.022

Lietaer, B., Ulanowicz, R. E., and Goerner, S. J. (2009). Options for managing asystemic bank crisis. Sapiens~2~(1), 1–16. Available At: https://journals.openedition.org/sapiens/747 (accessed on May 18, 2022).

Loginov, A. V. (2019). Second-order arguments, or do we still need tolerance in the public sphere? *CS&P.* 3, 319–332. doi:10.15826/csp.2019.3.4.080

Luzan, S. (2004). Regulation and management of enterprises with state participation: International experience. *Vopr. Ekon.* 9, 35–43. doi:10.32609/0042-8736-2004-9-35-43

Ma, J., and Jin, H. (2022). Increasing sustainability literacy for environmental design students: A transdisciplinary learning practice. Sustainability~14~(19),~12379.~doi:10.3390/su141912379

Makushkin, M. S., and Lapshin, V. A. (2021). Yield curve estimation in illiquid bond markets. *HSE. Econ. J.* 25 (2), 177–195. doi:10.17323/1813-8691-2021-25-2-177-195

Mendoza-Cavazos, Y. (2019). "Social welfare and sustainability," in *Encyclopedia of sustainability in higher education*. Editor W. Leal Filho (Cham: New York, USA: Springer). doi:10.1007/978-3-319-63951-2\_300-1

Morelli, J. (2011). Environmental sustainability: A definition for environmental professionals. J. Environ. Sustain. 1 (1), 1-10. doi:10.14448/jes.01

Naciti, V. (2019). Corporate governance and board of directors: The effect of a board composition on firm sustainability performance. *J. Clean. Prod.* 237, 117727. doi:10.1016/j.jclepro.2019.117727

Nazif Çatık, A., Huyugüzel Kışla, G., and Akdenız, C. (2020). Time-varying impact of oil prices on sectoral stock returns: Evidence from Turkey. *Resour. Policy* 69, 101845. doi:10.1016/j.resourpol.2020.101845

Nicolaou, C. (2021). "Development of business through the Internet and social media: The professional use of audiovisual media technologies through strategic tactics and practices," in *Handbook of research on IoT, digital transformation, and the future of global marketing* (London, UK: Business Science Reference), 193–211.

Noring, L. (2019). Public asset corporation: A new vehicle for urban regeneration and infrastructure finance.  $\it Cities~88,~125-135.~doi:10.1016/j.cities.2019.01.002$ 

Nurpeisova, A. A., Smailova, L. K., Akimova, B. Z., Borisova, E. V., and Niyazbekova, S. U. (2021). "Condition and prospects of innovative development

of the economy in Kazakhstan," in Socio-economic systems: Paradigms for the future. Studies in systems, decision and control. Editors E. G. Popkova, V. N. Ostrovskaya, and A. V. Bogoviz (Cham, Switzerland: Springer). doi:10.1007/978-3-030-56433-9 184

Ojekemi, O., Rjoub, H., Awosusi, A. A., and Agyekum, E. B. (2022). Toward a sustainable environment and economic growth in BRICS economies: Do innovation and globalization matter? *Environ. Sci. Pollut. Res.* 29, 57740–57757. doi:10.1007/s11356-022-19742-6

Petrucciani, S. (2020). "The critique of liberalism," in *The ideas of karl marx. Marx, engels, and marxisms* (Cham: Palgrave Macmillan). doi:10.1007/978-3-030-52351-0 2

Piscitelli, A., and D'Uggento, A. M. (2022). Do young people really engage in sustainable behaviors in their lifestyles? *Soc. Indic. Res.* 163, 1467–1485. doi:10.1007/s11205-022-02955-0

Pissourios, I. (2013). An interdisciplinary study on indicators: A comparative review of quality-of-life, macroeconomic, environmental, welfare and sustainability indicators. *Ecol. Indic.* 34, 420–427. doi:10.1016/j.ecolind.2013.06.008

Porreca, Z. (2020). Environmental sustainability and human capital development. Consilience 22, 48–57. doi:10.7916/consilience.vi22.6746

Qian, X., Chen, Y., and Cheung, E. S. (2019). "Changes and trend of public housing policy in Netherlands, Hong Kong and mainland China," in *Proceedings of the 2019 international conference on pedagogy, communication and sociology (ICPCS 2019)* (Paris: Advances in Social Science, Education and Humanities Research), 428–433. doi:10.2991/icpcs-19.2019.95

Rai, S. S., Rai, S., and Singh, N. K. (2021). Organizational resilience and social-economic sustainability: COVID-19 perspective. *Environ. Dev. Sustain.* 23, 12006–12023. doi:10.1007/s10668-020-01154-6

Rio Declaration on Environment and Development (1992). Rio de Janiero, Brazil: United Nations: Environment and Development. Available At: https://www.un.org/en/development/desa/population/migration/generalassembly/docs/globalcompact/A\_CONF.151\_26\_Vol.I\_Declaration.pdf (accessed on May 18, 2022). United Nations conference on the human environment.

Roje, G., and Redmayne, N. B. (2021). On the management and financial reporting for state assets – A comparative analysis between Croatia and New Zealand. *Public Money & Manag.* 41 (2), 118–126. doi:10.1080/09540962. 2020.1723261

Ruff, F. (2015). The advanced role of corporate foresight in innovation and strategic management – reflections on practical experiences from the automotive industry. *Technol. Forecast. Soc. Change* 101, 37–48. doi:10.1016/J.TECHFORE. 2014.07.013

Rustam, A., Wang, Y., and Zameer, H. (2020). Environmental awareness, firm sustainability exposure and green consumption behaviors. *J. Clean. Prod.* 268, 122016. doi:10.1016/j.jclepro.2020.122016

Sağlam, Y., and Egeli, H. A. (2018). A comparison of domestic demand and export-led growth strategies for European transition economies: Dynamic panel data analysis. *Foreign Trade Rev.* 53 (3), 156–173. doi:10.1177/0015732517734755

Schaltegger, S., Lüdeke-Freund, F., and Hansen, E. G. (2012). Business cases for sustainability: The role of business model innovation for corporate sustainability. *Int. J. Innovation Sustain. Dev.* 6 (2), 95–119. doi:10.1504/IIISD.2012.046944

Scott, J. T. (2013). The sustainable business: A practitioner's guide to achieving long-term profitability and competitiveness. Oxfordshire, UK: Routledge. doi:10. 4324/9781351276603

Semke, L.-M., and Tiberius, V. (2020). Corporate foresight and dynamic capabilities: An exploratory study. *Forecasting* 2, 180–193. doi:10.3390/forecast2020010

Shan, S., Ahmad, M., Tan, Z., Adebayo, T. S., Li, R. Y., and Kırıkkaleli, D. (2021). The role of energy prices and non-linear fiscal decentralization in limiting carbon emissions: Tracking environmental sustainability. *Energy* 234, 121243. doi:10.1016/j.energy.2021.121243

Shutaleva, A., Martyushev, N., Starostin, A., Salgiriev, A., Vlasova, O., Grinek, A., et al. (2022). Migration potential of students and development of human capital. *Educ. Sci. (Basel).* 12, 324. doi:10.3390/educsci12050324

Shutaleva, A., Nikonova, Z., Savchenko, I., and Martyushev, N. (2020). Environmental education for sustainable development in Russia. *Sustainability* 12 (18), 7742. doi:10.3390/su12187742

Shutaleva, A. V., Golysheva, M. V., Tsiplakova, Y. V., and Dudchik, A. Y. (2020). Media education and the formation of the legal culture of society. *P. Sci. Edu.* 45 (3), 10–22. doi:10.32744/pse.2020.3.1

Simangan, D., Sharifi, A., and Kaneko, S. (2021). Positive peace pillars and sustainability dimensions: An analytical framework. *Int. Stud. Rev.* 23 (4), 1884–1905. doi:10.1093/isr/viab054

Skvortsova, M. A., Zotova, E. V., Dragunova, I. V., and Malyasova, M. M. (2022). "Inflation volatility as a structural problem of sustainable development in pandemic conditions," in *Cooperation and sustainable development. Lecture notes in networks and systems.* Editors A. V. Bogoviz, A. E. Suglobov, A. N. Maloletko, and O. V. Kaurova (Cham New York, USA: Springer). doi:10.1007/978-3-030-77000-6\_140

So, A. Y. (2005). Beyond the logic of capital and the polarization model. *Crit. Asian Stud.* 37, 481–494. doi:10.1080/14672710500200573

Solovyov, M. M. (2008). Problems of assessing the effectiveness of state property management. *Manag. Russ. abroad* 4, 33–46.

Sørensen, A. (2022). Critical theory, immanent critique and neo-liberalism. Reply to critique raised in Copenhagen. *Philosophy Soc. Crit.* 48, 184–208. doi:10.1177/01914537211059506

Spangenberg, J. H. (2005). Economic sustainability of the economy: Concepts and indicators. *Int. J. Sustain. Dev.* 8 (1/2), 47–64. doi:10.1504/IJSD.2005.007374

Subjects of the quasi-public sector (2022). Concept, regulation. Available At: https://pravosite.kz/articles/post/1566-subekti-kvazigosudarstvennogo-sektora-ponyatie-normativnoe-regul (accessed on May 18, 2022).

Taubayev, A., Akenov, S. S., Ulybyshev, D. N., and Kernebaev, A. S. (2017). Institutional support of agro-industrial complex entities of quasi-public sector of Kazakhstan. *J. Adv. Res. Law Econ.* 8, 1350–1355.

Trencher, G., and Karvonen, A. (2017). Stretching "smart": Advancing health and well-being through the smart city agenda. *Local Environ.* 24 (7), 610–627. doi:10. 1080/13549839.2017.1360264

Valko, D. (2021). Environmental attitudes and contextual stimuli in the emerging environmental culture: An empirical study from Russia. SSRN J. 27, 2075–2089. doi:10.2139/ssrn.3730616

Valko, D. V. (2021). Impact of renewable energy and tax regulation on reducing greenhouse gas emissions in OECD countries: CS-ARDL approach. *Econ. Policy* 5, 40–61. doi:10.18288/1994-5124-2021-5-40-61

Vasiliev, V. P., and Sushko, V. A. (2021). Quality of life as an indicator of the effectiveness of public administration. *Vestn. Mosk. Univ. Ser. 18. Sociol. Politol.* 27 (4), 235–257. doi:10.24290/1029-3736-2021-27-4-235-257

Waiganjo, M., Godinic, D., and Bojan, O. (2021). Strategic planning and sustainable innovation during the COVID-19 pandemic: A literature review. *Int. J. Innovation Econ. Dev.* 7 (5), 52–59. doi:10.18775/ijied.1849-7551-7020.2015.75. 2005

WCED (1987). Our common future: Report of the world commission on environment and development. Oxford, USA: Oxford University Press for World Commission on Environment and Development. Available At: https://sustainabledevelopment.un.org/content/documents/5987our-common-future.pdf (accessed on May 18, 2022).

Wenzel, M. (2022). Taking the future more seriously: From corporate foresight to "Future-Making. *Acad. Manag. Perspect.* 36 (2), 845–850. doi:10.5465/amp.2020.

Yan, P., Wang, X., and Zhang, Q. (2020). Analysis on the efficient allocation of public assets and budget management based on evolutionary game theory. *Xit. Gongcheng Lilun yu Shijian/System Eng. Theory Pract.* 40 (11), 2872–2884.

Yizhe, X., Yan, C., Liu, H., Wang, J., Zhang, Y., and Yanlong, J. (2020). Smart energy systems: A critical review on design and operation optimization. Sustain. Cities Soc. 62, 102369. doi:10.1016/j.scs.2020.102369

Zeijl-Rozema, A. V., Cörvers, R., Kemp, R., and Martens, P. (2008). Governance for sustainable development: A framework. *Sust. Dev.* 16, 410–421. doi:10.1002/SD.367

Zhanbayev, R. A., Otyzbayeva, K. Zh., Zhanbayeva, L. A., Karbetova, Z. R., and Temirbaeva, G. R. (2021). Analysis of the scientific potential and possibilities of improving the quality of life in monotowns on the example of Arkalyk, Rudny and Zhezkazgan cities of the Republic of Kazakhstan. *Ugol'*. 11, 38–44. doi:10.18796/0041-5790-2021-11-38-44

Zhavoronkov, S. (2004). Inefficiency of state property management (the case of large enterprises). Vopr. Ekon. 9, 44–52. doi:10.32609/0042-8736-2004-9-44-52